\begin{table}%t1
%%\centering

\par
\caption{\label{samplet}The final sample.}
\begin{tabular}{ccc cc}
\hline\hline\noalign{\smallskip}
 {Name}& {Type}&$h/$kpc&$R_{\Omega}/$kpc & {Refs.}\\
\hline
NGC~247&HSB&$2.7$&$3.5\pm0.6$&$[4]$\\
NGC~300&HSB&$1.8$&$1.9\pm0.3$&$[4]$\\
NGC~2403&HSB&$2.1$&$1.8\pm0.3$&$[4]$\\
NGC~3109&HSB&$1.35$&$1.6\pm0.3$&$[4]$\\
NGC~3769&HSB&$1.5$&$2.2\pm0.8$&$[1]$, $[2]$\\
NGC~3877&HSB&$2.4$&$2.6\pm0.4$&$[1]$, $[2]$\\
NGC~3893&HSB&$2.0$&$1.7\pm0.4$&$[1]$, $[2]$\\
NGC~3972&HSB&$1.6$&$2.5\pm0.5$&$[1]$, $[2]$\\
NGC~3917&LSB&$2.6$&$2.6\pm0.2$&$[1]$, $[2]$\\
NGC~4010&LSB&$2.9$&$3.8\pm0.8$&$[1]$, $[2]$\\
NGC~4085&HSB&$1.3$&$1.9\pm0.3$&$[1]$, $[2]$\\
NGC~4157&HSB&$2.2$&$2.4\pm0.5$&$[1]$, $[2]$\\
NGC~4183&LSB&$2.65$&$2.9\pm0.8$&$[1]$, $[2]$\\
NGC~4217&HSB&$2.4$&$2.4\pm0.5$&$[1]$, $[2]$\\
NGC~4236&HSB&$2.7$&$3\pm0.4$&$[4]$\\
NGC~4389&HSB&$1.2$&$1.9\pm0.9$&$[1]$, $[2]$\\
UGC 128&LSB&$6.8$&$7\pm1.5$&$[3]$, $[5]$\\
UGC 1230&LSB&$4.5$&$6\pm1.2$&$[3]$, $[5]$\\
UGC 5005&LSB&$4.4$&$4.5\pm0.6$&$[3]$, $[5]$\\
UGC 5750&LSB&$5.6$&$5.8\pm1$&$[3]$, $[5]$\\
UGC 5999&LSB&$4.3$&$5\pm 1$&$[3]$, $[5]$\\
UGC 6399&LSB&$2.0$&$2.1\pm0.4$&$[1]$, $[2]$\\
UGC 6446&LSB&$1.5$&$1.6\pm0.6$&$[1]$, $[2]$\\
UGC 6667&LSB&$2.45$&$2.3\pm0.2$&$[1]$, $[2]$\\
UGC 6917&LSB&$2.4$&$2.3\pm0.2$&$[1]$, $[2]$\\
UGC 6923&LSB&$1.1$&$1.8\pm0.2$&$[1]$, $[2]$\\
UGC 6930&LSB&$2.0$&$2.4\pm0.3$&$[1]$, $[2]$\\
UGC 6962&HSB&$1.4$&$1.6\pm0.5$&$[1]$, $[2]$\\
UGC 6983&LSB&$2.2$&$1.9\pm0.5$&$[1]$, $[2]$\\
F 561-1&LSB&$2.8$&$2.8\pm0.5$&$[3]$, $[5]$\\
F 563-V2&LSB&$2.1$&$1.8\pm0.3$&$[7]$, $[6]$\\
F 565-V2&LSB&$2.7$&$3.5\pm0.5$&$[3]$, $[5]$\\
F 568-3&LSB&$4.0$&$3.1\pm0.5$&$[3]$, $[6]$\\
F 568-V1&LSB&$3.2$&$2.7\pm0.4$&$[3]$, $[6]$\\
F 571-V1&LSB&$3.2$&$4\pm1$&$[3]$, $[5]$\\
\hline
\end{tabular}
\tablebib{The references are coded as: $[1]$ Tully et~al. (\cite{Tully1996}), $[2]$ Verheijen \& Sancisi (\cite{Verheijen2001}), $[3]$ de Blok et~al. (\cite{de1995}), $[4]$ Kent (1987), $[5]$ de Blok \& McGaugh (\cite{de1997}), $[6]$ Swaters et~al. (\cite{Swaters2000}), $[7]$ McGaugh \& Bothun (\cite{McGaugh1994}).}
\end{table}